Response to Reviewers (Rebuttal Letter)

Eultor- Dr. Allilt I	Xasaue		
Journal Name- Jo	ournal of Current Pharma Research		
		Date-	
Dear Editor			
Subject- Submission of revised manuscript Number JCPF		JCPR- [Assigned	by the Journal
system]			
Thank you for yo	ur mail dated [Revision Date] enc	losing the reviewers comm	ents. We have
	the comments and revised the many		
•			Thomas to the
	ne reviewers comments point to p	omi in the below table, C	manges to the
	hlighted bold and coloured.		
Sr. No.	Correction Suggested	Response to	Reviewer
		comment	
1			
2			
	<u> </u>	L	
We hope the revis	sed version of the manuscript is su	itable for publication and le	ook forward to
hearing from you i	n due course.		
Your Name []		
A CC11: 41: 0 141: D	sainnatian F	1	
Affiliation with De	esignation []	